Author's response to reviews

Title: Understanding organisational development, sustainability, and diffusion of innovations within hospitals participating in a multilevel quality collaborative

Authors:

Michel LA Dückers (m.l.duckers@amc.uva.nl)
Cordula Wagner (c.wagner@nivel.nl)
Leti Vos (l.vos@lumc.nl)
Peter P Groenewegen (p.groenewegen@nivel.nl)

Version: 5 Date: 10 January 2011

Author's response to reviews:

Information is added to the methods: The research protocol for this study has been reviewed by the Medical Ethics Review Committee of the VU University Medical Centre (registered with the US Office of Human Research Protections as IRB00002991). The committee gave approval for the study to be performed. The study does not fall within the scope of the Medical Research Involving Human Subjects Act (WMO).